Author's response to reviews

Title: Food Access and Perceptions of the Community and Household Food Environment as Correlates of Fruit and Vegetable Intake among Rural Seniors

Authors:

Joseph R Sharkey (jrsharkey@srph.tamhsc.edu)
Cassandra M Johnson (cjohnson@srph.tamhsc.edu)
Wesley R Dean (wdean@srph.tamhsc.edu)

Version: 4 Date: 29 May 2010

Author's response to reviews: see over
Response to Production Team

We will address the requests by the Production Team.

Formatting changes to be made:
------------------------------

Major revisions (we require the author to make these changes)

1. Placement text: Please remove all figure and table placement text. All figures and tables must be cited in the manuscript in numerical order, so please check that they are all still cited in order once placement text is removed.

   RESPONSE: Placement text has been removed and listing has been checked.

2. References: reference 65, Submitted/ unpublished. Articles that have not been accepted for publication should not be cited in the reference list, unless they are publicly available as a pre-print. Instead, please refer to them in the text in the following way: (Authors: Title of manuscript, submitted).

   RESPONSE: The citation has been changed to identify reference 65 as being published.

3. Structure: Please add a Conclusion section to the manuscript.

   RESPONSE: Conclusion section added to the manuscript.

~

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

1. Email address: please include email addresses for all the authors on the title page, using the following format:
   AB: abcd@institution.ac.uk
   EF: efgh@generic.co.uk
   IJ: IJKL@corporation.com

   RESPONSE: Email addresses added

2. Structure: Please move tables and figure legends to after references.
RESPONSE: Tables and figure legends moved to after references.

3. References: the reference list should contain up to 30 authors' names for each citation. The term ?et al.? should not be used if there are fewer than 30 authors.

RESPONSE: reference list changed to include all authors.

4. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript.

RESPONSE: Manuscript reviewed.